



Article Traffic Stream Analysis by Radar Sensors on Two-Lane Roads for Free-Moving and Constrained Vehicles Identification

Giuseppe Cantisani^{1,*}, Giulia Del Serrone¹, Raffaele Mauro², Paolo Peluso¹, and Andrea Pompigna¹

- ¹ Department of Civil, Constructional and Environmental Engineering, University of Rome La Sapienza, Via Eudossiana 18, 00184 Rome, Italy
- ² Department of Civil, Environmental and Mechanical Engineering, University of Trento, Via Mesiano 77, 38123 Trento, Italy
- * Correspondence: giuseppe.cantisani@uniroma1.it

Abstract: This paper focuses on the analysis of traffic streams on two-lane highways, which are crucial components of transportation networks. Traffic flow measurement technologies, such as detection stations, radar guns, or video cameras, have been used over the years to detect the level of traffic and the operating conditions. This type of sensor can record a large amount of data which is useful to evaluate and monitor road traffic conditions, and it is possible to identify free-moving and constrained vehicles by processing the collected data. This study introduces an exponential headway model to identify the headway threshold above which vehicles can be considered as unconditioned. However, this value could identify vehicles which still retain some autonomy in their speed and maneuvering. Therefore, an additional criterion to distinguish between apparently and actually conditioned vehicles has been introduced, analyzing the speed differences between a vehicle and the preceding one. Three-month sequences of traffic monitored through radar devices placed on some Italian two-lane roads have been analyzed and an exponential headway model has been introduced, as an illustrative example. The results show that introducing the criterion of maneuvering freedom can significantly improve traffic flow analysis, modifying the starting critical values of 4 and 8 s per each studied section, to 2.5 and 3 s, approaching the values suggested by international manuals for traffic flow quality analysis.

Keywords: traffic radar sensors; two-lane roads; free-moving and constrained vehicles; leaderfollower influences; vehicle platoons; exponential headway model; apparently and actually conditioned vehicles

1. Introduction

Two-lane highways are a crucial part of the road network in many countries and are widely used for both systematic and non-systematic daily travel. Often, these roads make up the most extensive sub-network in a country's transportation system, enabling higher-level infrastructures, such as freeways, railways, ports, and airports, to be reached even in the most remote areas [1]. As urbanization spreads and traffic demand increases, two-lane highways face new challenges that require innovative and cost-effective solutions to improve operational and safety performance [2]. This goal can be achieved through the access to high-resolution data encompassing those recorded by the widespread of cost-effective traffic monitoring systems, such as radar sensors [3]. The traffic control units are local traffic controllers which generate a complete overview of traffic, weather and road conditions, using the sensors connected to it [4]. Advanced signal processing algorithms, indeed, provide detailed and precise data, regardless of the sensing technology employed (traffic sensors: inductive, sunk into the road, or microwave, installed above the road; video cameras: numberplate recognition and Closed-Circuit TeleVision (CCTV); weather sensors, etc.) [5]. In addition to the advanced technological aspect, these sensors are easy to install on the roadside, non-intrusive, high performance, low power, and provide a large amount



Citation: Cantisani, G.; Del Serrone, G.; Mauro, R.; Peluso, P.; Pompigna, A. Traffic Stream Analysis by Radar Sensors on Two-Lane Roads for Free-Moving and Constrained Vehicles Identification. *Sensors* 2023, 23, 6922. https://doi.org/10.3390/ s23156922

Academic Editor: Ram M. Narayanan

Received: 4 July 2023 Revised: 27 July 2023 Accepted: 1 August 2023 Published: 3 August 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). of information to be processed. These characteristics make them advantageous systems for monitoring the network. Thus, researchers can conduct comprehensive analyses of driver behavior in diverse traffic and road conditions over extended time periods.

One of the critical issues of two-lane highways is that passing occurs in the opposing lane of traffic, which has serious implications for traffic operation and safety. The limited passing opportunities result in a high level of interaction between vehicles, especially with the slow-moving ones. Therefore, it is essential to develop comprehensive models and metrics for performance analyses on this type of road, and to identify deficiencies and improve performances. Car-following models, within which each vehicle tries to maintain a desired time headway and speed behind the lead vehicle, attempt to simulate these interactions: drivers, in fact, continuously adjust their speeds based on their perceptions of the surroundings, the proximity to the desired speed, and the dynamics of the relationship with the lead vehicle [6]. The possibility of maintaining the desired speed depends on the presence of slower vehicles and on the option to overtake using the opposite lane. This interaction is particularly acute when both traffic flow and the percentage of slower vehicles increase, leading to vehicle platoons [7].

Understanding platooning behavior is crucial to modeling and analyzing traffic flow on two-lane highways accurately, as platoons can significantly affect operational performance and traffic flow. A platoon can be defined as a group of vehicles that travel together on the road, where following drivers adjust their speed, spacing, and acceleration according to the lead vehicle. As a result of these behavior dynamics, vehicles in the traffic stream can be grouped into two categories:

- A single vehicle or a platoon leader, which maintains a sufficiently large spatial and temporal headway with respect to the preceding vehicle, avoiding any influence from it and be able to travel at its desired or maximum speed at that moment.
- A platoon follower, which is unable to proceed at the desired speed and adjusts its driving behavior to follow the preceding vehicle (the platoon leader or another platoon component), while maintaining a reduced spatial and temporal detachment under its influence.

Passing lanes can mitigate the platoon impact on two-lane highways with high traffic levels, but they are not always available. For this reason, drivers may become frustrated and tend to accept smaller gaps in the opposing traffic to perform risky passing maneuvers, which can lead to head-on collisions. The average time that all vehicles delayed while traveling in platoons due to their inability to overtake was introduced in the 1985 edition of Highway Capacity Manual (HCM) [8] in terms of headways less than critical values. Even though threshold values have changed over the years and a new concept of Follower Density (FD) was introduced in the last edition [9], the threshold time headway for characterizing a vehicle in a non-free condition, i.e., constrained in a platoon, is a crucial element for analyzing two-lane roads according to the HCM.

Time headway is a common parameter for studying vehicle behavior in a single-lane traffic stream, and the threshold value is an extremely widespread approach to distinguish free-moving vehicles from platooned vehicles. As shown in Figure 1, the time headway ht_i of vehicle i + 1 is defined as the absolute passing time of vehicle i + 1 minus the absolute passing time of vehicle i, traveling in front of vehicle i + 1 (a); ht_i is the sum of the time gap gt_i and the occupation time ρ_i , where gt_i is the time between the rear of the lead vehicle and the front of the following vehicle and ρ_i is the time for the entire vehicle to pass the section. Headway can also be referred to in terms of distances between vehicles (b), and this is called spatial headway or spacing hs_i, which is the distance from the front of one vehicle to the front of the previous one; hs_i is equal to space gap gs_i in addition to the length of the vehicle l_i.



Figure 1. Time (a) and space (b) headways.

Different studies have proposed critical headway values to define vehicles that can travel at their desired speeds under free-flowing conditions. These vehicles are important for various applications in traditional traffic analyses and intelligent transportation systems [10,11]. Among all the works reviewed extensively in the following section, we can identify that a wide range of thresholds fall between 2 and 9 s. In this variety of works and values, we can distinguish two approaches to identify the threshold time headway in a single-line traffic stream, which refer to two different objectives and can be swayed toward two opposite extremes:

- The ones oriented toward identifying the Level Of Service (LOS) through the calculation of a Percentage of Time Spent Following (PTSF) in a platoon, the FD of vehicles in platoons, or even the Percentage of Followers (PF) in platoons. A not quite high value has to be taken, to avoid penalizing the analysis of traffic conditions by assigning the role of follower vehicles to those that are not actually impacted in their movements.
- The ones oriented toward defining Free-Flow Speed (FFS) or Operating Speed (V₈₅), setting a threshold to be used as a filter to screen conditioned vehicles and select only those that can be considered in free movement. In this case, the value must be not quite low, to avoid considering vehicles that are included within a platoon.

To date, the literature does not clearly distinguish between these two tendencies toward which the search for the threshold value for the analysis of platoon formation can deviate, according to the specific needs that guide the research. In this context, the purpose of this paper is to further investigate the role of the headway threshold and emphasize the significance of selecting an appropriate value, which may vary based on local circumstances (geographic location, territorial context, road geometry, general and specific regulations, driving habits, etc.) and the specific operational needs.

For these reasons, the paper introduces the use of low-cost traffic monitoring systems, specifically radar sensors, to gather high-resolution information on traffic behavior and provides insights into vehicle interactions on two-lane highways, enabling a deeper understanding of traffic flow characteristics and deficiencies. The paper introduces a two-fold approach to characterizing platoon formation using threshold headway values. By employing an exponential headway model, the study identifies headway thresholds that distinguish between constrained and non-constrained traffic situations useful for FFS or V85 analysis. The addition of an extra criterion, analyzing speed differences between vehicles, enhances the accuracy of identifying conditioned vehicles, and thus improves LOS analysis. The paper demonstrates the effectiveness of the proposed methodology by applying it to real-life traffic data collected from two study road sections in the Veneto Region of Italy. The practical implementation of the analysis process, including the identification of threshold values through statistical tests, validates the proposed approach's applicability in examining vehicle sequences.

The structure of the paper is as follows: Section 2 presents a comprehensive literature review on both road sensor technology for monitoring vehicle traffic and vehicular conditioning, summarizing the findings from the last 60 years of research on the headway threshold for free-moving vehicles. Section 3 outlines the methods used to characterize free-moving and conditioned vehicles and introduces the concepts of actual and apparent conditioning. Section 4 describes the research materials, including the characterization of vehicle traffic data using radar sensors, and identifies the specific test sections located in the Veneto Region of Italy. Section 5 provides a concrete example of the application of free-moving and constrained vehicles analysis, incorporating the headway threshold and maneuvering freedom criteria. Finally, Section 6 summarizes the study's conclusions.

2. Literature Review

2.1. Road Sensor Technology for Monitoring Vehicle Traffic

The management and monitoring activities of a road network require managers to collect data regarding the operational and actual conditions of the infrastructure. To date, scientific literature indicates that there are several data collection techniques, which can be briefly distinguished into static and dynamic ones [12]. The first technique uses devices installed in a fixed section of the road, and information, such as speed, volume, and traffic density, are collected for the entire vehicular flow [13,14]. In contrast, the second method consists mainly of probe vehicles, which have on-board GPS devices capable of sending real-time and high-frequency data about their motion [13]. It is shown that probe vehicles have the advantage of providing travel information along the entire road infrastructure; however, this information comes only from vehicles equipped with innovative devices, while the static technologies are limited to the survey section but provide information on the entire vehicular fleet [15,16].

2.2. Vehicular Conditioning

As stated in the Introduction section, a considerable amount of research has been conducted in the last 60 years to determine the threshold headway that identifies free-moving vehicles and separates constrained and non-constrained traffic situations for vehicles traveling on the same lane. Below, we present a chronological review of the main works that have addressed this issue, outlining the primary objectives of each study and the threshold values identified or suggested over time.

Miller (1961) [17] proposed a random queue model, using data from the Swedish State Roads Institute collected on a straight section of a two-lane road. The study concluded that free-flow conditions are established when the traffic flow rate is very low and vehicles can overtake freely, with a headway value of 8 s. When time intervals are less than 8 s, the study considered a second criterion based on the relative velocities between consecutive vehicles to determine constrained conditions in moving queues.

Wasielewski [18] examined a semi-Poisson time headway distribution model, suggesting a 6–8 s headway for two-lane roads and a 4 s headway for freeways.

Radwan and Kalevela (1985) [19] identified a headway greater than 6.5 s to classify non-following conditions between different classes of vehicles.

McLean (1989) [20] stated that drivers on two-lane roads are affected by the presence of the vehicle ahead of them for headways of 9 s or less.

Lam et al., (1990) [21] compared operating speeds on dry and wet pavements of twolane rural highways in New York state. They filtered free-moving vehicles considering vehicles with a minimum time gap of about 6 s or those heading a platoon of vehicles.

In 1991, Pursula and Enberg [22] conducted an analysis of speed-flow relationships, time headways, and platooning of traffic on two-lane rural roads in Finland. They used the platoon criterion of a 5 s time headway between successive vehicles for characteristics and level-of-service estimation.

Bennet et al., in 1994 [23] investigated critical headways at 58 study sites in New Zealand using different techniques to find the threshold. They identified a value between 3.0 and 4.5, recommending a value of 4.5 s to ensure that few following vehicles are misclassified as free.

Gattis et al., in 1997 [24] studied the behaviors of motorists. They considered a threshold headway value of 5 s between vehicles for platoon formation. They reviewed other studies by Messer and Morrall (1983) [25], Hoban (1983) [26], Guell and Virkler (1988) [27] in which the headway time used to define delay at two-lane highways varies from 3.5 s to 6 s and observed that a headway greater than 3 s may indicate a reduced desire to pass.

Dijker et al., in 1998 [28] analyzed differences in car following between congested and non-congested flows on Dutch roads and identified a critical gap value of 3.5 s for passenger cars and 5 s for trucks.

Vogel in 2002 [29] introduced a method that allows for establishing a threshold for free vehicles and determining how strongly the speed of the lead vehicle influences the speed of the following vehicle at a given headway in urban areas. The results suggested a time headway of 6 s as optimal for distinguishing between free and following vehicles for roads where traffic operates under capacity.

Van As in 2003 [30] selected 3.5 s as the threshold value to separate followers from platoon leaders on two-lane undivided roads in South Africa. Fifty percent of following vehicles have a headway of less than 3.5 s, 80% less than 4 s, and 99% less than 6 s.

Susarak et al., in 2004 [31] introduced platoon characteristics to measure traffic circulation quality on Tomey Express near Tokyo. They investigated new platoon criteria and platoon behavior and determined a critical headway relative speed method and exponential model. Based on the results on 18 sites, they defined a headway threshold of 3 s for passenger cars and 4 s for heavy vehicles.

In 2005, Fitzpatrick et al. [32] calculated free-flow speed at 79 tangent sites in suburban/urban and rural areas in six U.S. states by considering free-flowing vehicles as having at least a 5 s headway.

In 2005, Tseng et al. [33] determined the free-flow speed on Taiwan's rural and suburban highways by considering vehicles traveling with a headway greater than 5 s as vehicles in free-flow condition.

Figueroa and Tarko in 2005 [34] identified a headway greater than or equal to 5 s as the point at which vehicles travel in free-flow conditions on the tangent segment and horizontal curve, considering 158 spots on U.S. two-lane rural highway segments.

Al-Kaisy and Karjala in 2008 [35] presented an empirical investigation into performance indicators on two-lane rural highways using field data from four study sites in the state of Montana. They calculated the free-flow speed by averaging the speed of all free-moving vehicles traveling with headways greater than 8 s, and the percent of follower vehicles was calculated considering vehicles traveling with headways less than 3 s.

Al-Kaisy and Durbin in 2009 [36] investigated vehicular platoons on two-lane highways in Montana. The results showed that the increase in speeds is more notable at short headways, and it diminishes when headways reach a value in the range of 5 to 7 s.

Lay in 2009 [37] suggested that drivers are affected by the presence of the vehicle ahead for headways of 9 s or less. Thus, he argued that the critical headway is still a matter of some debate with values quoted between 2.5 and 9 s, and with a best estimate of 4 s, and most in the range from 3.0 to 4.5 s.

Al-Kaisy and Karjala in 2010 [10] examined car-following interaction and the definition of free-moving vehicles on two-lane rural highways at eight study sites in Montana State. Empirical observations suggest that the car-following interactions on two-lane rural highways generally cease beyond a headway value of 6 s.

Hashim in 2011 [38] analyzed speed characteristics on two-lane rural highways in Egypt and investigated the relationship between 85th percentile speed and headways. A headway value of 5 s revealed the threshold between constrained vehicles and free-moving vehicles.

Lobo et al., in 2011 [39] conducted a study on Portuguese two-lane rural roads to define a gap value between two successive vehicles from which the vehicles can be considered as traveling in a non-platoon condition. They found a free gap of 6 s and highlighted the possible differences that may occur from site to site as a reflection of road geometry and from area to area (city, state, or country) as a function of general driver behavior.

In 2013, Semeida [40] posited that vehicles travel in free-flow conditions with a time headway of at least 8 s when developing new models to evaluate the level of service and capacity of rural multi-lane highways in Egypt.

Robertson et al., in 2014 [41] collected and studied traffic from different sites on freeways and multilane highways within the state of Texas for Level of Service models. Analyzing data from radar devices, they considered no free-flow conditions if the headway is 5 s or greater.

In 2015, Penmetsa et al. [42] studied two-lane intercity highways under mixed traffic conditions in India. It was assumed that vehicles traveling in the same lane with a relative speed of 2 km/h or less were in the car-following model. The gap corresponding to a 50% probability of not following was chosen as the critical gap, which was found to be 2.6 s; a following probability of about 80% considers vehicles with a headway of less than 4 s and 100% less than 5.5 s.

Boora et al., in 2018 [43] identified a headway value of 10 s for free-flow conditions. They introduced a headway and relative speed distribution analysis to define a second criterion for following behavior based on the speed difference between consecutive vehicles. A range of -7 to +15 km/h for the speed difference between the leader and follower was defined by data to identify followers and non-followers among vehicles with a headway of less than 10 s. Then, considering an acceptance curve method and a probability of 50%, a critical headway of 4.2 s was indicated.

Saha et al., in 2019 [44] investigated platoon phenomena on two-lane highways with heterogeneous traffic in northeast India to establish a headway threshold for free-moving vehicles. Their investigation indicated a value of about 6 s under which vehicles form platoons and that few drivers move in a platoon by choice even if they have a gap of more than 6 s and passing opportunities.

Al-Kaisy et al., in 2019 [45] presented an empirical investigation into car-following interactions from 15 study sites in Idaho, Montana, and Oregon. They argued that more aggressive drivers may start to interact with the lead vehicle and adjust their speeds when they are very close to the lead vehicle, while on the other hand, more cautious drivers may start to interact with the lead vehicle and adjust their speeds at a relatively large distance from the lead vehicle. Values for the critical headway are expected in a range that includes aggressive and cautious drivers, are site-specific, with a lower limit generally between 1 and 2 s and an upper limit between 6 and 7 s. Calculating the number of following vehicles in the whole transition range, they identified the cut-off value for platooning as the value to which an equivalent number of vehicles with smaller headways corresponds. Cut-off values from 15 sites range between 1.98 s and 3.85 s, and the author finally indicated a comprehensive critical headway value for identifying a vehicle in a following status as 2.5 s. The research work by Al-Kaisy et al. [45] was included in the project NCHRP 17-65-Improved Analysis of Two-Lane Highway Capacity and Operational Performance [25], a project approved by the TRB Committee on Highway Capacity and Quality of Service (ACP40) in 2019 for inclusion in the HCM.

Figure 2 depicts a graphical representation of the critical headway values for the platoon formation threshold identified in the reviewed past studies, ranging from a minimum of 2.5 s to a maximum of 9 s. The graph shows the survival function trend based on the literature values, which is the probability trend of finding values greater than or equal to a certain headway in the examined literature works. The continuous function is the trend of the survival function of the normal distribution that fits the data, with a mean of 5.29 s and a standard deviation of 1.68 s.



Figure 2. Survival function of the critical headway values for the platoon formation threshold identified in the reviewed studies: sampled and normal fitted trends.

At this point, it is crucial to consider the various editions of the HCM and their distinct characteristics concerning the formation and impact of platoons on traffic quality and Level of Service determination for two-lane roads. The 1985 edition of the HCM [8] introduced the concept of Percent Time Delay (PTD), which stands for the percentage of time all vehicles are delayed while traveling in platoons due to their inability to overtake. However, obtaining real-time PTD measurements is impractical; therefore, the percentage of vehicles that travel behind a platoon leader at speeds lower than their desired speed and at headways less than a critical value is often used as a surrogate measure to determine the percentage of delays. The initial threshold introduced by the HCM was 5 s, which was maintained in subsequent editions of 1994 [46] and 1997 [47] guidelines for two-lane roads. In the 2000 edition of the HCM [48], PTD was renamed PTSF, which represents the freedom to maneuver, comfort, and convenience of travel by quantifying the percentage of travel time that vehicles must travel in platoons behind slower vehicles due to the inability to pass. The surrogate measure for PTSF is the percentage of vehicles traveling with headways less than a threshold. In this case, the suggested critical value was lowered to 3 s. This critical threshold value of 3 s remained unchanged in subsequent editions of 2010 [49] and 2016 [50], until the latest edition of 2022 [9], which presents a new methodological framework to analyze the quality of service of two-lane highways based on the Al-Kaisy et al. [45] and NCHRP 17-65 [25] studies. The HCM 2022 methodology [9] introduced FD as a new service measure, which is described as the number of vehicles in a follower state per mile per lane. FD overcomes the previous issues with PTSF performance measure since it can be directly measured in situ. In this context, a follower vehicle is defined by a headway equal to or less than 2.5 s, and FD is the service measure for all two-lane highway configurations.

3. Methods: Characterization of Free-Moving and Conditioned Vehicles

3.1. Main Aspect of Vehicular Conditioning and Exponential Headway Model

As mentioned earlier, single-lane traffic flow vehicles can be classified as either:

Free-moving, unconstrained, or unconditioned vehicles, which are not in a platoon
or are at the head of it and can travel at their desired speed with enough space to
maneuver within the traffic flow;

• Constrained or conditioned vehicles within a platoon that must adjust their speed to that of the vehicle in front of them, potentially traveling at a slower speed than the desired one and are unable to overtake.

The headway, which is the time interval between two vehicles, is a common parameter used to characterize the two types of vehicles on a two-lane road. An often-used approach is to identify a critical time interval as the threshold for separating a conditioning situation from a non-conditioning one. There are two main types of models that have been used to represent single-lane time headway distributions: Simple models consisting of a single statistical distribution of observed headways, and models that are a set of two distributions, representing headways of constrained and free-moving vehicles in appropriate proportions [11,51].

The main simple models for temporal headway representation include Negative Exponential, Translated Exponential, Erlang, and Lognormal distributions [11,51,52]. Among these, the Negative Exponential distribution is suitable for modeling low-density traffic streams distributed without platoons along the road axis. In this case, the maximum degree of randomness is associated with the arrival of vehicles. If we consider a vehicle arrival process in a monitoring section that meets the following conditions:

- 1. Absence of memory: the number of arrivals in any time interval is independent of that in any other time interval.
- 2. Ordinariness: the probability of two or more arrivals occurring in a small-time interval is negligible compared to the probability of a single arrival occurring.
- 3. Stationarity: the arrival intensity λ is constant over time.

the probability that m arrivals will occur in a time interval Δt follows a Poisson vehicle arrival process, see Equation (1):

$$P_m = \frac{a^m}{m!} e^{-a} \tag{1}$$

where:

 $m = 0, 1, 2, \ldots, n, \ldots$

$$\lambda \Delta t$$
 (2)

If Equation (1) is fulfilled, the inter-arrival times between arrivals are exponentially distributed with a probability density function given by Equation (3):

a =

$$f(\tau) = \lambda e^{-\lambda \tau} \tag{3}$$

However, the Negative Exponential distribution provides non-zero probability values even for very small, almost zero, headways, which contradicts the fact that vehicles have non-zero lengths. To address this inconsistency, it is possible to assume the translated exponential as the probabilistic model for headways, which includes the smallest possible headway value under the examined traffic conditions [11,51,53].

In the presence of even small headways characterized by low frequencies, as well as an increase in mutual interferences between vehicles, an Erlang distribution can be assumed, which is a special case of the Pearson type III distribution [54]. The Erlang distribution is the probabilistic model for the inter-temporal distance between the instants of two non-contiguous events, generally the (n)th and (n + k)th, of a Palm Event Flow (Limited Memory Flow, a generalization of a Stationary Poisson Process) [55–58]. However, the Erlang distribution has always been used in traffic engineering for temporal vehicle spacings, which are intervals between successive contiguous and non-contiguous passes. With appropriate choices of the k-parameter values, it is often possible to achieve good fits of the Erlang distribution to experimental traffic data obtained by observing various discharge regimes [11,54,58,59].

In a dense traffic stream, in which most of the inter-vehicular time between vehicles falls within a limited number of amplitude classes, a log-normal probability density function can realistically be assumed to interpret the headway between vehicles [11,60].

While simple models consider the full experimental range of headways, they do not provide as broad a view of the factors influencing the pattern of arrivals. However, they usually require from the data fewer parameters for estimations. Mixed models, such as Buckley's Semi-Poisson Model [61], Cowan's M3 and M4 models [52], and Branston's Generalized Queuing Model [62], result from a random arrival pattern modification, on the assumption that the two physical factors that prevent traffic in a single lane from behaving as a random phenomenon are the constraints imposed by stability and safety considerations and the constraints imposed by the lack of overtaking opportunities.

The distribution resulting from the modification takes the general form for mixed models, as in Equation (4):

$$f(\tau) = \vartheta g(\tau) + (1 - \vartheta) h(\tau)$$
(4)

where $f(\tau)$ is the probability density function of all headways, $g(\tau)$ is the probability density function of following headways, $h(\tau)$ is the probability density function of free-moving headways, ϑ is the proportion of following vehicles, and $(1 - \vartheta)$ is the proportion of freemoving vehicles. Other alternative methods include a data-driven approach that achieves greater adaptability to experimental data by interpreting statistically emerging behaviors from real data without the need for a priori assumptions or probability laws for speeds and time headways, such as the Two-Lane Road Statistical Platooning Model (TLR-SPM) [2].

3.2. Actual and Apparent Conditioning

The exponential headway model assumes that the headways of free-flowing traffic or headways between each platoon group tend to distribute randomly and can be modeled as a negative exponential headway distribution. The three key points of this model are:

- If vehicle arrivals in the selected section are totally random, and the time spacing between them is distributed according to an exponential random variable, the counting process is Poissonian.
- The exponential model for headways, like the Poisson model for counts, conforms to traffic counts when traffic flow is low.
- If the traffic flow cannot be considered low, the arrivals are no longer random, and the headways no longer follow an exponential distribution.

According to the exponential model, free-moving vehicles have a distribution of interarrival times that seems to be consistent with the distribution of random arrivals. Thus, the lowest value for which the headway distribution is no longer approximated by a translated exponential represents the minimum headway for which vehicles no longer exhibit the characteristics of unconditioned vehicles. Since the hypothesis of arrival randomness cannot be rejected for inter-arrival time values equal to or greater than this minimum headway, the absence of conditioning with respect to the preceding vehicle cannot be rejected either.

If the above conditions are not met, i.e., for inter-arrival time values less than the minimum headway, according to the exponential model, we can say that arrivals are characterized by a lack of complete randomness, and thus we cannot consider them as free-moving. We can then say that, according to the exponential model, these vehicles appear as conditioned. If conditioning is expressed as a lack of randomness, it indicates the existence of some regularity in the arrivals at the examined section, which results in some dependence or uniformity of motion characteristics between a given vehicle and the one in front of it.

However, it should be noted that the exponential model is based on an aggregate analysis of headways and their distributions. It is interesting to know what occurs when the individual vehicle passage has a headway below the threshold identified in the aggregate analysis. According to the exponential model, it is no longer a free-moving vehicle, and subsequently could be designated as conditioned. This leads to another query: can we classify it as actually conditioned merely since it has a headway below the threshold and assume that this situation explicitly expresses the existence of an authentic effect on the vehicle's movement by the preceding vehicle? This cannot be said with certainty. The vehicle, in fact, may arrive at the monitoring section with a headway below the threshold, but it still retains some maneuvering freedom over the preceding vehicle, and thus it can manage its speed autonomously without actual conditioning. Therefore, it is appropriate to refer to this phenomenon as apparent conditioning since it cannot be identified considering only the headways. Therefore, an additional criterion to observe no evidence of maneuvering freedom is necessary to assert that the vehicle appears to be actually conditioned. To sum up, a conditioned vehicle has a headway below a certain threshold, which consequently does not conform to the hypothesis of completely random arrival according to the exponential model; therefore, there are two different situations:

- An apparent conditioning situation, where the vehicle still has some maneuvering freedom compared to the preceding vehicle.
- An actual conditioning situation, where the vehicle exhibits a lack of maneuvering freedom compared to the preceding one. In the latter case, the vehicle tends to conform to the kinematic conditions of the preceding vehicle, resulting in the creation of a platoon.

3.3. Conditioning by Necessity and by Choice

The primary circumstances leading to actual conditioning and the formation of platoons include:

- The presence of traffic congestion.
- The presence of a slower leading vehicle.
- High density of opposing traffic, preventing overtaking.
- Specific prohibitions on overtaking.

In addition to these situations, which can be identified as actual conditioning by necessity, there are also instances in which a driver may spontaneously choose to assume a position of uniformity with respect to the vehicle in front of them. In these scenarios, the driver may maintain a specific following distance and speed that matches the vehicle in front of them. This can result in a platoon of vehicles with a more uniform speed and inter-vehicle spacing, known as following by choice or intentional platooning [10,43]. It is based on the driver's autonomous decision and not on external factors, such as traffic congestion/interactions or overtaking impracticality/prohibition. Situations where drivers choose to closely follow another vehicle can occur when:

- Their desired speed is very close to that of the leader of the platoon.
- They conform to a specific driving style dictated by the road geometry or environmental conditions.

In general, a distinction between following by choice and following by necessity cannot be made by only analyzing the data of headways and maneuvering freedom (for example, by examining the relationship between the kinematic characteristics of two consecutive vehicles) available from traffic monitoring. Therefore, a more comprehensive analysis of traffic conditions and driving behavior is necessary to fully understand the reasons behind the formation of platoons and the spacing between vehicles.

However, we can differentiate between choice and necessity in terms of spatial spacing (referred to as spacing for brevity). In following-by-necessity situations, there may be a greater inclination for vehicles to maintain smaller inter-vehicle spacing in the hope of overtaking a slower vehicle as soon as possible, resulting in a more tightly spaced platoon of vehicles. Conversely, in following-by-choice situations, vehicles may maintain a larger inter-vehicle spacing and form a more loosely spaced platoon. Although the vehicles in these platoons tend to conform to a common driving style, resulting in a certain regularity of arrival times (conditioning) and reduced freedom of maneuver (effective conditioning), they do not necessarily influence each other.

It is worth noting that a larger inter-vehicle spacing may also be observed between vehicles in apparent conditioning, where the headway is below the threshold of the expo-

nential model, but there is still some freedom of maneuver. In these circumstances, due to driving styles or strategies, drivers may temporarily maintain a larger safety buffer between their vehicle and the preceding vehicle, resulting in a more loosely spaced platoon.

4. Materials: Characterization of Vehicle Traffic Data by Radar Sensors for the Analysis of Vehicle Conditioning

The objective of this study bases the proposed methodology on the analysis and processing of traffic data acquired through a static method; the following sections detail the adopted data sample.

4.1. Survey Method and Traffic Data

Vehicles' identification in the free-moving and constrained conditions has been carried out by analyzing data collected through the Automatic Statistical Traffic Detection System of the Italian national road network manager, ANAS SpA [63]. The ANAS Road network currently has an automatic statistical traffic detection system with about 1200 counting sections. The sensors used for counting and classifying vehicles and measuring traffic flows are based on advanced microwave technology. They are designed to be installed directly over the center of the monitored lane. Their low power draw makes them highly versatile in use since they can be powered by a photovoltaic panel as well as directly by the mains (230 V/50 Hz). The sensors work in both directions of travel and can thus detect vehicles traveling against the flow: it can therefore continue working in the case of overtaking, lane changes, open worksites, etc. A single cable connects the sensor to the local controller. The main technical specifications of the adopted sensors are listed in Table 1:

Table 1. Technical specifications of the adopted sensors.

Technology	Microwave	
Operating Conditions	Temperature: -20 °C to $+60$ °C Humidity: 0% to 100%, without	
Power Consumption	condensation 12 VDC to 24 VDC 1 W approx.	
Output Data	Number, direction, speed, length, occupation time, headway and gap between vehicles, vehicle category.	
Data Generated	Time stamp in milliseconds Sensor diagnostics	
Dimensions And Weight	$165 \times 95 \times 280 \text{ mm} (\text{H} \times \text{W} \times \text{D})$ 1.8 kg without accessories	
Enclosure Material	ABS Stainless-steel roof	
Protection Rating	IP66	
Mounting System	Collar/locking assembly	
Mounting	Height: 6 m to 7.5 m Position: aligned with center of lane	

The sensors in the counting stations send data to a centralized monitoring system called PANAMA (Piattaforma ANAs per il Monitoraggio e l'Analisi), which then provides verification, processing, and validation of the acquired data through a series of computerized control processes. In addition, the ability of the sensors to detect and capture the actual passing through the survey section is evaluated by ANAS personnel through simultaneous overlapping observation of the control unit readings and videos of the vehicles passing through [64]. Each control unit provides a sample of detailed traffic data, of which an extract of the reported information is shown in Table 2:

Time Reference	Thousandths of a Second	Lane	Direction	Speed (km/h)	Time Gap (s)	Headway (s)	Vehicle Class
01/02/2019 00:00:40	200	2	А	72	97.1	97.3	2
01/02/2019 00:01:12	300	2	А	76	31.9	32.1	2
01/02/2019 00:01:35	0	2	А	79	22.6	22.7	4
01/02/2019 00:01:36	900	2	А	90	1.8	1.9	2
01/02/2019 00:02:01	100	1	D	69	172.1	172.3	2
01/02/2019 00:03:32	300	1	D	92	91.1	91.3	7
01/02/2019 00:04:14	100	2	А	69	157.2	157.4	2
01/02/2019 00:04:33	100	2	А	86	18.8	18.9	2

Table 2. Example of control unit traffic survey.

For each detected data, it is possible to know the following information: Time reference, which is the time instant of data acquisition (day and hour); Thousandths of a second, which is the time that should be added to the Time reference to determine the time of the actual passage of the vehicle; Lane, which is the lane on which the vehicle travels; Direction, which indicates the ascending or descending direction of travel; Speed, which is the value of the recorded speed (km/h); Time Gap, which is the time in a given section between the passage of the rear of a passing vehicle and the head of the following vehicle (s); Headway, which is instead the time in a given section between the passage of the head of the following vehicle (s); and Vehicle Class, which is the Famas 9 + 1 classification code to identify different types of vehicles. The classification scheme adopted by Famas is: 1 Motorcycle, 2 Car, 3 Car + trailer, 4 Van, 5 Truck < 7.5 m, 6 Truck > 7.5 m, 7 Truck-trailer, 8 Articulated, 9 Bus, 10 Other.

The PANAMA database is constantly growing with real-time data collected from various continuous monitoring stations. In this paper, a portion of this database is used for illustrative purposes, limited to two monitored sections for 3 months, with a detailed number of observations in terms of individual vehicle transits as follows.

4.2. Identification and Characterization of Test Sections

The proposed survey was carried out on the highway's road network of the Veneto region managed by ANAS. In detail, the national roads n.14 and n.309 and the related traffic control units, Control Unit 218 (CU 218) and Control Unit 1333 (CU 1333), respectively, have been considered. The extraction covers the months of February, May, and August of a normal operating year to account for the different conditions determined by seasonal traffic variation cycles. Table 3 shows the number of the recorded observations for each control unit in the examined time periods:

Months	CU 218	CU 1333
February	226,613	265,321
May	269,708	300,520
August	271,787	311,512
Total	768,108	877,353

 Table 3. Number of recorded observations by the control units.

The control units will be explained in detail below in terms of their location and the infrastructural and spatial context in which they are installed.

4.2.1. Control Unit 218 (CU 218)—National Road n.14 (SS14)

The coordinates of the control unit 218, installed along the SS14, are given in the WGS84 reference system (lat. 45.59°, long. 12.47°). Figure 3a shows the installation site, which is a straight trunk, where it is assumed that the user behavior is not strongly influenced by the horizontal geometry of the infrastructure itself. At the monitoring point,

there is a lateral emergency stopping bay situated adjacent to the travel lane. The satellite image in Figure 3b shows that the CU 218 is installed in a non-urbanized area with low residential density with predominantly agricultural activities. The surrounding 500 m radius area is characterized by few accesses to private areas and an intersection with a rural path, justified by the agricultural use of the local land.



Figure 3. Control unit 218 on the national road n.14: (**a**) installation site with two lanes and lateral emergency stopping bay; (**b**) surrounding 500 m radius area (Google Earth).

4.2.2. Control Unit 1333 (CU 1333)—National Road n.309 (SS309)

The coordinates of the control unit 1333, installed along the SS309, are given in the WGS84 reference system (lat. 45.19°, long. 12.27°). Figure 4a shows the installation site, which is even in this case a straight trunk, where it is assumed that the user behavior is not strongly influenced by the horizontal geometry of the infrastructure itself. The satellite image in Figure 4b shows that the CU 1333 is installed in an industrialized and agricultural area with medium residential density. The surrounding 500 m radius area is characterized by various accesses to different industrial activities, heralding a great influence of heavy vehicles on traffic operating conditions.



Figure 4. Control unit 1333 on the national road n.309: (**a**) installation site with two lanes; (**b**) surrounding 500 m radius area (Google Earth).

5. Results: Free-Moving and Constrained Vehicles Analysis

5.1. Headway Threshold Criterion for the Detection of Free-Moving Vehicles

In Section 3, we have observed the exponential model, which suggests that the arrivals of unconditioned vehicles can be described as Poissonian. Additionally, their inter-arrival time distribution follows a negative exponential distribution. When analyzing the experimental data described in Section 5.1, we can use the threshold headway criterion to identify free-flowing vehicles. This involves scrutinizing the inter-arrival times of the sampled vehicles in each monitoring section and identifying the lowest value for which the sample distribution can be approximated by an exponential distribution. This value is deemed as the threshold for the absence of conditioning: any inter-arrival values higher than the

threshold are thus consistent with the distribution of random arrivals. Correspondingly, the associated sample headway distribution is considered as consistent with the distribution of random arrivals.

From a statistical standpoint, the threshold time value represents the minimum headway for which the hypothesis of absence of conditioning with respect to the preceding vehicle, or equivalently the randomness of arrival, cannot be rejected. In general, it can be observed that if $f(\tau)$ is an exponential probability density function and $F(\tau)$ is the related cumulative distribution function, the logarithmic transformation $L(\tau)$ of the complement to one of the $F(\tau)$ has a linear trend. Thus, to identify the critical headway, we can analyze the logarithmic transformation of the complement to one of the experimental headway distribution function $\tilde{F}(\tau)$, i.e., $\tilde{L}(\tau) = \ln \left[1 - \tilde{F}(\tau)\right]$, and we can examine the minimum

time interval value τ^* , for which $L(\tau)$ can be approximated by a straight line.

An explanatory example of how this can be carried out can be provided by using the traffic data collected according to the methods discussed in Section 5.1, which relate to the two monitoring sections presented in Section 5.2. Only values less than 300 s were considered for the two monitoring sections, under the assumption that arrivals with headways greater than this value were entirely random and not part of the same traffic stream. By grouping headway values less than 5 min into bins of 1 s (in order that the 0 s bin includes headways ≤ 0.5 s, the 1 s bin includes headways between 0.5 and 1.5 s, etc.), we can evaluate the experimental probability density function $\tilde{f}(\tau)$, the experimental probability distribution function $\tilde{F}(\tau)$, and the logarithmic transformation of its complement to one $\tilde{L}(\tau)$.

For each of the two considered monitoring sections, i.e., 218 and 1333, and directions of travel, after grouping the headways less than 300 s into bins of 1 s, we proceeded to test progressively every value of possible critical headway τ * between 0 and 9 s. For every τ *, $\tilde{F}(\tau)$ was computed for τ * $\leq \tau \leq 300$ s. Thus, we analyzed the trend of $\tilde{L}(\tau)$ and estimated the regression line of the {X = τ ; Y = (τ)}. Upon analyzing the obtained trends, Figure 5 shows that as τ * increases, the possibility of approximating $\tilde{L}(\tau)$ with a straight line significantly improves, with a Goodness of Fit (GoF) that can be initially observed merely graphically.



Figure 5. Cont.



Figure 5. Experimental trend of $L(\tau)$ for $\tau \ge \hat{\tau}*$ with $\hat{\tau}* \le \tau \le 300$ s and $0 \le \hat{\tau}* \le 9$ s: (a) control unit 218; (b) control unit 1333.

However, a graphical assessment of linearity as in Figure 5 does not allow us to identify the threshold value τ^* as we intended. Measures of GoF, such as SSE (Sum of Squared Residuals) and R2 (Coefficient of Determination), were evaluated by considering \sim

the experimental trend of $L(\tau)$ for $\tau \ge \hat{\tau}$ * and that of the regression line. Additionally, the regression line obtained from each $\hat{\tau}$ * was employed to estimate the theoretical frequencies of each headway class for 1 s bins between 0 and 30 s. By considering the deviations between the sample frequencies and the theoretical frequencies obtained with the regression, MAPE (Mean Absolute Percentage Error) and MXAPE (Maximum Absolute Percentage Error) were also evaluated. The values for both sections, varying with τ^* , are reported in Table 4:

•	Section 218				Section 1333					
τ*	$\mathbf{N}(au \geq \hat{ au^*})$	R2	SSE	MAPE	MXAPE	$N(\tau \ge \tau^*)$	R2	SSE	MAPE	MXAPE
0	363,375	0.9578	0.2026	5.22%	28.80%	441,177	0.9486	1.5639	17.54%	49.83%
1	362,114	0.9806	0.0733	3.11%	19.91%	440,176	0.9579	1.0497	14.39%	47.49%
2	294,183	0.9943	0.0178	1.70%	9.76%	343,460	0.9705	0.5978	11.59%	36.30%
3	242,699	0.9978	0.0060	1.06%	5.47%	233,076	0.9775	0.3781	9.73%	27.78%
4	219,188	0.9990	0.0024	0.71%	3.37%	174,223	0.9812	0.2650	8.50%	22.48%
5	204,321	0.9995	0.0011	0.51%	2.21%	140,319	0.9836	0.1963	7.58%	19.05%
6	193,053	0.9997	0.0006	0.38%	1.45%	117,540	0.9852	0.1493	6.78%	16.58%
7	183,559	0.9998	0.0003	0.30%	1.06%	100,536	0.9865	0.1148	6.06%	14.84%
8	175,411	0.9999	0.0002	0.24%	0.80%	87,319	0.9877	0.0877	5.39%	13.45%
9	167,998	0.9999	0.0001	0.21%	0.55%	76,490	0.9890	0.0657	4.76%	12.25%

Table 4. GoF measures for linear regression of the {X = τ ; Y = (τ)} values for sections 218 and 1333.

Based on these clear trends, however, we are not yet able to identify which $\hat{\tau}$ we can assume as the threshold value. It is necessary to develop a specific criterion that can be used

to select the minimum value of $\hat{\tau}$ * that makes the linear approximation of $L(\tau)$ acceptable, and thus estimate the suitable $\tau * = \tau *$ to represent the threshold value between freeflow and congested traffic. A criterion based on a statistical GoF test between probability distributions was introduced to address this question. The Kolmogorov–Smirnov (KS) test, commonly found in literature [65], was used to compare a sample distribution with a reference distribution (in this case, exponential). The KS test statistic D_n, in Equation (5):

$$D_n = \max \left| \widetilde{F}(\tau) - F(\tau) \right| \tag{5}$$

is obtained by comparing the experimental probability distribution functions, i.e., each sample distribution $\widetilde{F}(\tau | \hat{\tau}^* \leq \tau \leq 300 \text{ s})$ with $0 \leq \hat{\tau}^* \leq 9$, and the theoretical probability distribution function with which the comparison is made, $F(\tau)$, at each point τ of the related sample. The null hypothesis H_0 is that $\widetilde{F}(\tau) = F(\tau)$. If the maximum difference between the theoretical and observed frequencies is large, we reject the null hypothesis for large values of D_n , while we do not reject it for small values of D_n . The critical value of D_n (which we can denote as KSC) is defined for a certain sample size N and level of significance α , i.e., $KSC = KSC_{\alpha,N}$.

If $D_n \leq KSC_{\alpha,N}$, then this result leads to the non-rejection of the null hypothesis H_0 , which states that the data come from the hypothesized distribution model, with the probability distribution function $F(\tau)$. Usually, a significance level of 5% ($\alpha = 0.05$) is adopted as the reference value. In general, GoF tests, such as the KS test, provide some information about the fit between the sample and the theoretical model, based on *p*-values and critical values. In the case of the KS test, lower values of the statistic, particularly below the critical value, indicate an agreement between the sample and the theoretical model. While the meaning of the D_n statistic is intuitively evident, calculating its probability distribution (under the null hypothesis H_0) is more complicated. It is shown that under the null hypothesis H_0 , the probability distribution of the test statistic D_n does not depend on the functional form of $F(\tau)$, and therefore the critical values can be obtained from specific statistical tables for small samples or calculated based on simple formulas for large samples, as a function of the significance level α . In general, the formula for $KSC_{\alpha,N}$ is:

$$KSC_{\alpha,N} = \frac{\sqrt{-0.5\ln\left(\frac{\alpha}{2}\right)}}{\sqrt{N}} \tag{6}$$

and if we consider the typical value of $\alpha = 0.05$, then $KSC_{\alpha,N} = \frac{1.36}{\sqrt{N}}$. However, in general, this type of test is extremely sensitive to the sample size. As the sample size increases, the *p*-values decrease dramatically, even in cases where the good agreement of data with the candidate model is evident [66]. As for $KSC_{\alpha=0.05,N}$, we see that it decreases rapidly as N increases. Considering the available sample sizes (in the order of several hundred thousand for each section), the KS test would lead to rejecting the exponential distribution for all critical headway values, even if they are sufficiently large, due to its intrinsic limitation. Despite the clear linear trend of $\widetilde{L}(\tau)$ for $\tau \geq \tau *$ as $\tau *$ increases, due to the sample size N and the reduction in critical values $KSC_{\alpha=0.05,N}$, the KS test (as well as other similar GoF tests) provides extremely low *p*-values (and D_n values much larger than excessively low

critical values). This would always make the null hypothesis H_0 of agreement with any

theoretical model unacceptable [67–70]. In formulating the criterion to find the appropriate value of $\hat{\tau}$ * that can be assumed as the threshold value τ *, the problem is to make correct inference even in the case of a large N, by always applying the KS test to test the null hypothesis of exponential distribution of vehicle interarrival times greater than $\hat{\tau}$ *. In literature, there are implementations of KS tests on large samples, including headways, by randomly extracting a large number of small sub-samples from the original sample [67,71]. Thus, in defining the criterion for choosing the threshold for free-moving vehicles, we have adopted this approach, based on random resampling without replacement from the initial sample of size N, by extracting a large number m of small sub-samples of size n \ll N.

As previously mentioned, the choice to work with a big number (m = 1000) of small random sub-samples (n = 300) compared to the hundreds of thousands of headways in the original samples solves the problem of GoF test oversensitivity, related to the loss of power of hypothesis tests for very large sample sizes, that leads to rejecting significant distributions. Therefore, for each of the 1000 sub-samples, the D_n statistic was calculated, and to evaluate the acceptability of H₀ (exponential distribution) for each test threshold value τ^* based on $\hat{\tau}_*$, a significance level of $\alpha = 0.05$ was chosen resulting in a critical value of $KSC_{\alpha=0.05,n=300} = 0.0748$. The proposed criterion for assessing the overall GoF involves calculating the D_n statistic for each sub-sample and obtaining the average D_n of these values. This criterion serves as a heuristic guideline to address the oversensitivity of traditional KS tests when dealing with large datasets. By comparing the mean value D_n with a critical value, it provides a basis for accepting or rejecting the null hypothesis for the entire dataset. Regarding the aforementioned KS tests, Table 5 presents the mean value D_n of the D_n statistic over 1000 random sub-samples of size 300 for two illustrative cases. It also indicates (**) that the null hypothesis is not rejected based on the heuristic criterion $D_n < KSC_{\alpha=0.05,n=300}$. By considering the smallest value of $\hat{\tau}$ * between 0 and 9 with a $D_n < KSC_{\alpha=0.05,n=300}$, the critical headway value τ^* is identified. Thus, using the exponential model in the two road sections for test, a threshold headway τ^* of 4 s is identified in 218 and 8 s in 1333. Therefore, we can say that vehicles with a headway greater than or equal to these values appear in the sample with characteristics that allow us to attribute their arrivals to a degree of randomness compatible with a Poisson process. As a result, we can consider these as free-moving vehicles. Conversely, vehicles with headways less than these threshold values are not characterized by randomness of arrivals, and thus may express a certain conditioning.

Table 5. KS test for section with H₀ considering $F(\tau | \tau * \leq \tau \leq 300 \text{ s})$ with $0 \leq \tau * \leq 9$ compatible with an exponential distribution for sections 218 and 1333.

Section 218			Section 1333			
τ*	\overline{D}_n	$D_n < KSC_{\alpha=0.05, n=300} (=no;^{**}=yes)$	\overline{D}_n	$D_n < KSC_{\alpha=0.05, n=300}$ (= <i>no</i> ;**= <i>yes</i>)		
0	0.2599		0.1667			
1	0.2187		0.2407			
2	0.1687		0.2091			
3	0.0956		0.1400			
4	0.0443	**	0.1212			
5	0.0711	**	0.1293			
6	0.0655	**	0.1325			
7	0.0698	**	0.0933			
8	0.0636	**	0.0777	**		
9	0.0738	**	0.0642	**		

5.2. Speeds and Manoeuvring Freedom Criterion for the Detection of Actual/Apparent Conditioning

We have highlighted that this status can be classified as actual conditioning if vehicles show a reduction in maneuvering freedom with a tendency to conform to the kinematic conditions of the preceding vehicle. On the other hand, it can be classified as apparent conditioning if the vehicles still retain a certain degree of maneuvering freedom compared to the previous vehicle. As previously stated, we need to further characterize this conditioning by introducing some other criteria that allow us to distinguish between actual conditioning and apparent conditioning. In order to ensure the accuracy of the analysis, a criterion that considers only the degree of maneuvering freedom of each vehicle must be established. Based on the information available for each passage reported in Section 5.2, this criterion can be derived from the analysis of the speed of each vehicle and its relative speed with respect to the preceding vehicle.

An initial analysis that helps to differentiate between free-moving vehicles and conditioned vehicles concerns the experimental bivariate distribution of headway and speed. This is shown in Figure 6 as a heat map for headways less than or equal to 60 s and speeds less than or equal to 130 km/h. From the experimental bivariate distributions of speed and headway, we can observe a separation between two distinct zones in the heat maps, which can be identified by the threshold between non-free (sub-sample 2) and free-moving vehicles (sub-sample 3). Below this threshold (i.e., 4 s for the CU 218 and 8 s for the CU 1333)



the speed distribution covers a wider range that also includes values of 30–40 km/h, which are infrequent for headways above the same threshold.

Figure 6. Experimental bivariate distribution of headway and speed-heat map for headways ≤ 60 s and speeds ≤ 130 km/h: (a) road section 218; (b) road section 1333. Three samples can be identified: (1) entire sample, (2) non-free vehicles, and (3) free-moving vehicles.

The threshold headway criterion, by itself, is insufficient in providing a comprehensive characterization of conditioning situations since it fails to distinguish between actual and apparent conditioning. Nevertheless, it facilitates the identification of unconditioned vehicles, and thus the threshold headway criterion establishes a value with a sufficient level of caution in filtering only fully and completely free-moving vehicles. Thus, free-moving vehicles in sub-sample 3 in Figure 6 can be filtered from the overall data using the headway threshold and further analyzed to determine speed percentiles in free-flow traffic conditions. For example, as introduced in the Introduction section, this can be useful in the analysis of operational speed (i.e., the 85th percentile of operating speed distributions or V_{85}), that is a critical factor for road safety, since it significantly influences the frequency and severity of accidents [72,73], and it is widely acknowledged as a benchmark value for evaluating consistency in homogeneous road sections [74–76]. Figure 7 illustrates the experimental and best-fitting distributions (Gumbel Max for CU 218 and Chi-Squared for CU 1333) and the corresponding V_{85} values in the two monitored sections from the speed data in sub-sample 3.

After distinguishing the two subgroups of free-moving and conditioned vehicles based on the threshold headway and identifying their differences in terms of speed distributions, further analyses can be conducted in relation to the relative speeds. The speed difference between a vehicle and the preceding one can be used as a parameter to represent the freedom of maneuver, where a lower absolute value implies a higher likelihood of the vehicle being influenced by the preceding one and vice versa. Figure 8 presents representative histograms of the experimental distributions of speed differences for the overall sample, as well as the two partitions of conditioned vehicles (sub-sample 2) and free-moving vehicles (sub-sample 3).



Figure 7. Sample and fitted cumulative distribution function for vehicle speeds within free-moving sub-samples: (**a**) road section 218; (**b**) road section 1333.



Figure 8. Cont.



Figure 8. Experimental distributions of speed differences for the overall sample, sub-sample 2, and sub-sample 3: (a) road section 218; (b) road section 1333.

The figure also includes the distribution that best fits the experimental data based on the KS test. As evidenced by the images and numerical results in Table 6, the distributions appear substantially different in terms of both descriptive statistics and theoretical distributions that accurately describe their trends.

Table 6. Descriptive statistics and theoretical distributions for speed difference
--

Section Group	Overall Data	Section 218 Sub-Sample 2	Sub-Sample 3	Overall Data	Section 1333 Sub-Sample 2	Sub-Sample 3
Mean	$-2.0299 imes 10^{-4}$	0.31696	-0.25052	$-2.17 imes10^{-4}$	0.10065	-0.40323
Std. Deviation	15.157	5.6813	18.7S	11.59	9.296	18.042
Skewness	-0.06348	0.54154	-0.00456	0.1003	-0.32462	0.35945
Kurtosis	3.649	18.288	1.5158	13.244	19.841	4.4177
Fitting Function (best)	Laplace	Laplace	Logistic	Laplace	Laplace	Logistic

The descriptive statistics indicate that the mean values are close to zero in all cases, but the standard deviation values differ among the samples, with sub-sample 2 having values closer to the mean than the other two samples. The skewness values are close to zero, indicating that the distributions are approximately symmetrical. However, all kurtosis values are greater than 3, suggesting that all the distributions are more peaked than a normal distribution, with sub-sample 2 having a more peaked distribution than the other samples. These differences in kurtosis values suggest structural differences in the data distribution, primarily in comparison to sub-sample 3. These observations apply to both the CU 218 and CU 1333 test sections illustrated in Figure 8, which displays the histograms and the best-fit function.

The selection of the Logistic distribution for sub-sample 3 and the Laplace distribution for both sub-sample 2 and the overall data further indicates differences in the headway distribution between sub-samples 2 and 3, in terms of the presence of extreme values and the probability of producing values close to the mean. Therefore, speed differences in sub-sample 3 with headway values equal to or greater than the threshold value may be more prone to extreme variations than those in sub-sample 2.

Following Miller's (1961) proposal [17], later adopted by Boora et al., (2018) [43], we can utilize these differences in distributions to identify the range of speed differences in which the frequencies of the non-free-moving vehicle sub-sample exceed those of the free-moving vehicle one. If the speed difference distribution of sub-sample 2 exceeds that of sub-sample 3 within a certain interval, then there are more vehicles in sub-sample 2 with speed differences falling within that interval than there are free-moving vehicles

in sub-sample 3. This suggests that vehicles in sub-sample 2 with speed differences in that interval may have probabilistically different behaviors from free-moving vehicles in sub-sample 3. We can consider the vehicles in sub-sample 2 with speed differences falling within the interval that characterizes probabilistically different behaviors from free-moving ones as actually conditioned, and those with differences outside the interval as apparently conditioned.

Using this approach, as shown in Figure 9, we can identify the range of speed differences in which the frequencies of the distribution of conditioned situations (actual + apparent) surpass those of free-moving ones, representing a range of conditioning prevalence that can indicate actual conditioning. Based on this range, we can define a new criterion to represent the freedom of maneuver and to distinguish between these two situations:

- Vehicles with speed differences outside this range of conditioning prevalence are apparently conditioned;
- Vehicles with speed differences within this range of conditioning prevalence are actually conditioned.



Figure 9. Experimental distributions of speed differences for the overall sample, sub-sample 2, sub-sample 3, and conditioning prevalence interval: (**a**) road section 218; (**b**) road section 1333.

In the case under consideration, the ranges that can be identified based on the experimental distributions, as shown in Figure 9, consider speed differences between -5 km/h and 5 km/h for road section 218 and between 6 km/h and 6 km/h for road section 1333.

5.3. Headway Acceptance Curves in Vehicle Conditioning

Therefore, we can consider that a vehicle is actually conditioned if it demonstrates:

- A headway value below the threshold identified with the exponential model (in the example, 4 s for CU 218 and 8 s for CU 1333).
- A speed difference with the preceding vehicle within the range of conditioning prevalence, identified based on the comparison between the experimental frequencies of speed differences among the conditioned sample (actual + apparent) and the unconditioned sub-sample (in the example, [-5 km/h, 5 km/h] for the road section 218 and [-6 km/h, 6 km/h] for the road section 1333).

Thus, an apparently conditioned vehicle is a vehicle that, despite passing through the monitoring section with a headway lower than the threshold value, appears to have a reduced level of randomness, yet still exhibits a certain degree of maneuverability. This maneuverability is evident in the speed difference with the preceding vehicle, which lies outside the range of conditioning prevalence. By applying both criteria to the entire monitoring sequence, we can identify actual conditioning vehicles and analyze their percentage trend according to the headway class. In Figure 10, for CU 218, we can observe a decrease from 100% at 0 s headway to 90% at 1 s, 80% at 2 s, 65% at 3 s, and ultimately 0% at the threshold value of 4 s. Similarly, for CU 1333, we can see a decrease from 100% at 0 s to 80% at 2 s and progressively to 62% at 4 s, 48% at 6 s, 23% at 7 s, and ultimately 0% at the threshold value of 8 s.



Figure 10. Histograms for the experimental density distribution of actual conditioning vehicles for headway class below the exponential model threshold: (**a**) road section 218; (**b**) road section 1333.

As suggested by Boora et al. [43], we can plot the cumulative percentage of apparently conditioned vehicles, representing the cumulative probability of non-following vehicles. Figure 11 shows the results for the two test sections. For CU 218, a headway threshold value of 3 s includes approximately 60% of the apparently conditioned vehicles, while a threshold value of 2 s reduces this percentage to 40%. Similarly, for CU 1333, a threshold value of 6 s includes approximately 80% of the apparently conditioned vehicles, while values of 3 s and 2 s correspond to 50% and 35%, respectively. Configured as an acceptance curve [43], this plot can be used to identify a critical gap value, above or below which vehicles would travel as free or following vehicles, respectively. Considering the 50% probability value for each of the two cases, we obtain a critical value of slightly under 2.5 s for CU 218, and about 3 s for CU 1333. Despite differences in the threshold headway values of the exponential model, this range appears narrower and comparable to the threshold indicated by the latest editions of the HCM for determining measures of effectiveness (PTSF or FD) for analyzing traffic quality and Levels of Service.

5.4. Further Insights on the Distribution of Inter-Vehicle Distance in Vehicle Conditioning

An interesting aspect to investigate is certainly the spacing between vehicles, namely, the spatial distance between two consecutive vehicles. In Table 7, the trends of the average spacing (in meters) between consecutive vehicles for headway bins between 0 ($\tau \le 0.5$ s) and 10 ($9.5 \le \tau < 10.5$ s) are shown. As we can see, the average spacing for the threshold headway is 87 m for CU 218 ($\tau^* = 4$ s) and 157 m for CU 1333 ($\tau^* = 8$ s). Since we consider vehicles with headways greater than t* as free moving, they correspond to average headways greater than 87 m for CU 218 and 157 m for CU 1333.

If we shift our attention to analyzing the probability distribution of the spacing for vehicles below the threshold that we defined as conditioned simply due to the fact that they do not meet the criteria of the exponential model for free-moving vehicles (and thus composed of a combination of vehicles that are apparently and actually conditioned), we observe that the average spacing is 34.2 m, with a 90th percentile of 59 m for CU 218 (sample distribution and lognormal fitting in Figure 12a), and 47 m, with a 90th percentile of 99 m for CU 1333 (sample distribution and lognormal fitting in Figure 12b).



Figure 11. Experimental cumulative distribution of apparently conditioned vehicles with respect to the total number of conditioned vehicles (actual + apparent): (**a**) road section 218; (**b**) road section 1333.

Table 7. Average spacing (m) for headway bins (1 s) for sections 218 and 1333.

Headway Bin	Average S	pacing (m)
(s)	Section 218	Section 1333
0 (0–0.5)	7	8
1 (0.5–1.5)	21	19
2 (1.5–2.5)	39	33
3 (2.5–3.5)	63	51
4 (3.5–4.5)	87	71
5 (4.5–5.5)	111	92
6 (5.5–6.5)	135	114
7 (6.5–7.5)	159	136
8 (7.5–8.5)	183	157
9 (8.5–9.5)	206	178
10 (9.5–10.5)	230	199



Figure 12. Histograms and lognormal fitting for spacing (m) in conditioned (actual + apparent) vehicles: (a) road section 218; (b) road section 1333.

If we consider the two groups into which we can divide the conditioned vehicles, i.e., actually and apparently conditioned, based on the prevalence interval of the conditioning concerning speed differences, the analysis of distributions clearly highlights differences between the two clusters. In CU 218, the average spacing is 45.4 m for apparently conditioned vehicles, and it is reduced to 31.8 m for actually conditioned vehicles, with a standard

deviation decreasing from 19.7 m to 15.4 m. In CU 1333, the average spacing is 64.4 m for apparently conditioned vehicles, and it is reduced to 41.2 m for actually conditioned vehicles, with a standard deviation decreasing from 40.2 m to 29.0 m. Consequently, we can confirm that the two groups exhibit statistically different behaviors, even in terms of spatial distancing.

Figure 13 presents the histograms for the two groups, i.e., apparently and actually conditioned, of spatial distances and the resulting distributions from the best approximation according to the usual KS test (lognormal for actually conditioned and beta for apparently conditioned in CU 218; 3-parameter lognormal for actually conditioned and gamma for apparently conditioned in CU 1333).



Figure 13. (a) Road section 218; (b) road section 1333.

6. Conclusions

Two-lane highways are essential components of transportation systems in many countries, but they face operational and safety challenges due to limited passing opportunities and platooning behavior. The increasing availability of low-cost traffic monitoring systems, such as radar sensors, provides high-resolution information in various traffic and road geometry conditions, allowing for comprehensive analyses of vehicle behavior on this specific type of road. This enables a deeper analysis of vehicular interactions on two-lane highways, which is essential for identifying deficiencies and improving operational performance for this important category of roads.

The literature review highlighted the importance of threshold headway values, ranging from 2.5 to 9 s, in characterizing platoon formation and differentiating between constrained and non-constrained traffic situations. The exponential headway model helps to identify the headway threshold above which vehicles can be considered unconditioned due to the randomness of their arrivals, and statistical tests were used to determine the critical

value. However, the threshold headway value, determined using the exponential model, could identify vehicles which still retain some autonomy in their speed and maneuvering. Therefore, the paper introduced as a novelty an additional criterion to distinguish between apparently and actually conditioned vehicles: the analysis of speed differences between a vehicle and the preceding one. Using this range as a criterion, a distinction was established between actual conditioning and apparent conditioning. This approach allowed for a more accurate assessment of traffic flow quality, without relying solely on the exponential model and avoiding the excessively cautious approach that may result from it.

The application of the proposed methodology to real-life sequences from two study road sections in the Veneto Region of Italy demonstrated the randomness of arrival thresholds in a wide range of values, which in the two specific cases were 4 s and 8 s, respectively and significantly differed from the thresholds of the most recent editions of the HCM, which is 2.5 s. Introducing the criterion of maneuvering freedom, critical values for platoon formation within a narrower range could be found. In fact, through the identification of headway acceptance curves for actual/apparent conditioning, critical headway turned out to be 2.5 s and 3 s, approaching the values suggested by international manuals for traffic flow quality analysis.

The findings of this study provided valuable insights into the characterization of vehicular conditioning on two-lane highways. Although the number of road sections studied in this article was limited, the findings served as an exploratory analysis of the proposed methodology. Future research could further investigate this methodology by collecting data using radar detection systems in a greater number of sections. These research opportunities could involve distinguishing local traffic and infrastructure situations to provide additional and in-depth contributions to the development of effective strategies for improving road safety and operational performance.

Author Contributions: Conceptualization, G.C. and R.M.; methodology, R.M. and A.P.; software, A.P., G.D.S. and P.P.; validation, A.P.; formal analysis, A.P. and G.D.S.; investigation, A.P. and P.P.; data curation, P.P.; writing—original draft preparation, A.P.; writing—review and editing, G.D.S.; supervision, G.C. and R.M.; project administration, G.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available on request from the corresponding author. The data are not publicly available due to confidentiality issues and respect for privacy.

Acknowledgments: We thank the national public company ANAS for making its Traffic Observatory information available and for the technical support.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Al-Kaisy, A. Two-Lane Highways: Indispensable Rural Mobility. Encyclopedia 2022, 2, 625–631. [CrossRef]
- Mauro, R.; Pompigna, A. A Statistically Based Model for the Characterization of Vehicle Interactions and Vehicle Platoons Formation on Two-Lane Roads. *Sustainability* 2022, 14, 4714. [CrossRef]
- Cantisani, G.; Del Serrone, G.; Peluso, P. Reliability of Historical Car Data for Operating Speed Analysis along Road Networks. Sci 2022, 4, 18. [CrossRef]
- 4. Tyburski, R. A review of road sensor technology for monitoring vehicle traffic. Inst. Transp. Eng. J. 1988, 59, 27–29.
- 5. Klein, L. Sensor Technologies and Data Requirements for ITS Applications; Artech House: Boston, MA, USA, 2001.
- Saifuzzaman, M.; Zheng, Z. Incorporating Human-Factors in Car-Following Models: A Review of Recent Developments and Research Needs. *Transp. Res. Part C Emerg. Technol.* 2014, 48, 379–403. [CrossRef]
- Del Serrone, G.; Cantisani, G.; Grilli, R.; Peluso, P. Effectiveness of Climbing Lanes for Slow-Moving Vehicles When Riding Uphill: A Microsimulation Study. *Vehicles* 2023, 5, 744–760. [CrossRef]

- 8. Transportation Research Board; National Research Council. *Highway Capacity Manual*; Transportation Research Board: Washington, DC, USA; National Research Council: Washington, DC, USA, 1985.
- 9. Transportation Research Board; National Research Council. *Highway Capacity Manual*; Transportation Research Board: Washington, DC, USA; National Research Council: Washington, DC, USA, 2022.
- 10. Al-Kaisy, A.; Karjala, S. Car-Following Interaction and the Definition of Free-Moving Vehicles on Two-Lane Rural Highways. *J. Transp. Eng.* **2010**, *136*, 925–931. [CrossRef]
- 11. Guerrieri, M.; Mauro, R. A Concise Introduction to Traffic Engineering; Springer: Berlin, Germany, 2021.
- 12. Del Serrone, G.; Cantisani, G.; Peluso, P. Speed Data Collection Methods: A Review. *Transp. Res. Procedia* 2023, 69, 512–519. [CrossRef]
- 13. Ajmar, A.; Arco, E.; Boccardo, P.; Perez, F. Floating Car Data (FCD) for Mobility Applications. *Int. Arch. Photogramm. Remote Sens. Spatial Inf. Sci.* **2019**, 42, 1517–1523. [CrossRef]
- 14. Chase, R.T.; Williams, B.M.; Rouphail, N.M.; Kim, S. Comparative Evaluation of Reported Speeds from Corresponding Fixed-Point and Probe-Based Detection Systems. *Transp. Res. Rec. J. Transp. Res.* **2012**, 2308, 110–119. [CrossRef]
- Findley, D.J.; Cunningham, C.M.; Hummer, J.E. Comparison of Mobile and Manual Data Collection for Roadway Components. *Transp. Res. Part C Emerg. Technol.* 2011, 19, 521–540. [CrossRef]
- Prabha, R.; Kabadi, M.G. Overview of Data Collection Methods for Intelligent Transportation Systems. Int. J. Eng. Sci. IJES 2016, 5, 16–20.
- 17. Miller, A.J. A Queueing Model for Road Traffic Flow. J. R. Stat. Soc. Ser. B 1961, 23, 64–75. [CrossRef]
- 18. Wasielewski, P. Integral Equation for the Semi-Poisson Headway Distribution Model. Transp. Sci. 1974, 8, 237–247. [CrossRef]
- 19. Radwan, A.E.; Kalevela, S. Investigation of the effect of change in vehicular characteristics on highway capacity and level of service. *Transp. Res. Rec.* **1985**, *1005*, 65–71.
- 20. McLean, J.R. Two-Lane Highway Traffic Operations: Theory and Practice; Gordon & Breach: London, UK; New York, NY, USA, 1989.
- 21. Lamm, R.; Choueiri, E.M.; Mailaender, T. Comparison of operating speeds on dry and wet pavements of two-lane rural highways. *Transp. Res. Rec.* **1990**, *1280*, 199–207.
- Pursula, M.; Enberg, A. Characteristics and level-of-service estimation of traffic flow on two-lane rural roads in Finland. *Transp. Res. Rec.* 1991, 1320, 135–143.
- 23. Bennett, C.R.; Dunn, R.C.M. Critical Headways on Two-Lane Highways in New Zealand. In Proceedings of the Second International Symposium on Highway Capacity, Sydney, Australia, 9–13 August 1994; Volume 1, pp. 91–100.
- Gattis, J.L.; Alguire, M.S.; Townsend, K.; Rao, S. Rural Two-Lane Passing Headways and Platooning. *Transp. Res. Rec.* 1997, 1579, 27–34. [CrossRef]
- Washburn, S.; Al-Kaisy, A.; Luttinen, T.; Dowling, R.; Watson, D.; Jafari, A.; Bian, Z.; Elias, A. Improved Analysis of Two-Lane Highway Capacity and Operational Performance; Final Report Project NCHRP 17-65, National Cooperative Highway Research Program, Web-Only Document, No. 255; Transportation Research Board: Washington, DC, USA, 2018. [CrossRef]
- 26. Hoban, C.J. Towards a Review of the Concept of Level of Service for Two-Lane Roads. Aust. Road Res. 1983, 13, 216–218.
- 27. Guell, D.L.; Virkler, M.R. Capacity Analysis of Two-Lane Highways. Transp. Res. Rec. 1998, 1194, 199–203.
- Dijker, T.; Bovy, P.H.; Vermijs, R.G. Car-Following under Congested Conditions: Empirical Findings. *Transp. Res. Rec.* 1998, 1644, 20–28. [CrossRef]
- 29. Vogel, K. What Characterizes a "Free Vehicle" in an Urban Area? *Transp. Res. Part F Traffic Psychol. Behav.* 2002, 5, 15–29. [CrossRef]
- 30. Van As, C. The Development of an Analysis Method for the Determination of Level of Service of Two-Lane Undivided Highways in South Africa; Project Summary; South African National Roads Agency Limited: Pretoria, South Africa, 2003.
- 31. Surasak, T.; Okura, I.; Nakamura, F. Measuring level of services on multi-lane expressways by platoon parameters. *J. Infrastruct. Plan. Man. JSCE* **2004**, 2004, 23–32. [CrossRef] [PubMed]
- Fitzpatrick, K.; Miaou, S.P.; Brewer, M.; Carlson, P.; Wooldridge, M.D. Exploration of the Relationship between Operating Speed and Roadway Features on Tangent Sections. J. Transp. Eng. 2005, 131, 261–269. [CrossRef]
- Tseng, P.; Lin, F.; Shieh, S.-L. Estimation of Free-Flow Speeds for Multilane Rural and Suburban Highways. *Transportation* 2005, 6, 1484–1495.
- Medina, A.M.F.; Tarko, A.P. Speed Factors on Two-Lane Rural Highways in Free-Flow Conditions. *Transp. Res. Rec.* 2005, 1912, 39–46. [CrossRef]
- Al-Kaisy, A.; Karjala, S. Indicators of performance on two-lane rural highways: Empirical investigation. *Transp. Res. Rec.* 2008, 2071, 87–97. [CrossRef]
- Al-Kaisy, A.; Durbin, C. Platooning on Two-Lane Two-Way Highways: An Empirical Investigation. J. Adv. Transp. 2009, 43, 71–88.
 [CrossRef]
- 37. Lay, M.G. Handbook of Road Technology, 4th ed.; Spon Press: New York, NY, USA, 2009.
- Hashim, I.H. Analysis of speed characteristics for rural two-lane roads: A field study from Minoufiya Governorate, Egypt. *Ain Shams Eng. J.* 2011, 2, 43–52. [CrossRef]
- Lobo, A.; Jacques, M.; Rodrigues, C.; Couto, A. Free-gap evaluation for two-lane rural highways. *Transp. Res. Rec.* 2011, 2223, 9–17. [CrossRef]

- Semeida, A.M. New models to evaluate the level of service and capacity for rural multi-lane highways in Egypt. *Alex. Eng. J.* 2013, 52, 455–466. [CrossRef]
- 41. Robertson, J.; Fitzpatrick, K.; Park, E.S.; Iragavarapu, V. Determining level of service on freeways and multilane highways with higher speeds. *Transp. Res. Rec.* **2014**, 2461, 85–93. [CrossRef]
- Penmetsa, P.; Ghosh, I.; Chandra, S. Evaluation of Performance Measures for Two-Lane Intercity Highways under Mixed Traffic Conditions. J. Transp. Eng. 2015, 141, 04015021. [CrossRef]
- 43. Boora, A.; Ghosh, I.; Chandra, S.; Rani, K. Measurement of free-flow conditions on multilane intercity highways under heterogeneous traffic conditions. J. S. Afr. Inst. Civ. Eng. 2018, 60, 2–9. [CrossRef]
- 44. Saha, P.; Roy, R.; Sarkar, A.K.; Pal, M. Preferred Time Headway of Drivers on Two-Lane Highways with Heterogeneous Traffic. *Transp. Lett.* **2019**, *11*, 200–207. [CrossRef]
- 45. Al-Kaisy, A.; Jafari, A.; Washburn, S. Following Status and Percent Followers on Two-Lane Highways: Empirical Investigation. *Civ. Eng. Res. J.* 2019, 7, 61–70. [CrossRef]
- 46. Transportation Research Board; National Research Council. *Highway Capacity Manual*; Transportation Research Board: Washington, DC, USA; National Research Council: Washington, DC, USA, 1994.
- 47. Transportation Research Board; National Research Council. *Highway Capacity Manual*; Transportation Research Board: Washington, DC, USA; National Research Council: Washington, DC, USA, 1997.
- 48. Transportation Research Board; National Research Council. *Highway Capacity Manual*; Transportation Research Board: Washington, DC, USA; National Research Council: Washington, DC, USA, 2020.
- 49. Transportation Research Board; National Research Council. *Highway Capacity Manual*; Transportation Research Board: Washington, DC, USA; National Research Council: Washington, DC, USA, 2010.
- 50. Transportation Research Board; National Research Council. *Highway Capacity Manual*; Transportation Research Board: Washington, DC, USA; National Research Council: Washington, DC, USA, 2016.
- 51. Mauro, R. Traffic and Random Processes; Springer: Cham, Switzerland, 2015; pp. 1–119. [CrossRef]
- 52. Cowan, R.J. Useful Headway Models. Transp. Res. 1975, 9, 371–375. [CrossRef]
- 53. Abuelenin, S.M.; Abul-Magd, A.Y. Effect of Minimum Headway Distance on Connectivity of VANETs. *AEU-Int. J. Electron. Commun.* 2015, 69, 867–871. [CrossRef]
- 54. Ben-Edigbe, J.; Rahman, R.; Mashros, N. Research Article Headway Distributions Based on Empirical Erlang and Pearson Type III Time Methods Compared. *Res. J. Appl. Sci. Eng. Technol.* **2014**, *7*, 4410–4414. [CrossRef]
- 55. Haight, F.A. The Generalized Poisson Distribution. Ann. Inst. Stat. Math. 1959, 11, 101–105. [CrossRef]
- 56. Wentzel, E.; Ocharov, L. Applied Problems in Probability Theory; Mir Publishers: Moscow, Russia, 1986.
- 57. Gartner, N.; Messer, C.J.; Rathi, A.K. *Traffic Flow Theory: A State-of-the-Art Report*, 1st ed.; Committee on Traffic Flow Theory and Characteristics, TRB: Washington, DC, USA, 2001.
- Mauro, R.; Branco, F.; Guerrieri, M. Contribution to the Platoon Distribution Analysis in Steady-State Traffic Conditions. *Periodica Polytech. Civ. Eng.* 2014, 58, 217–227. [CrossRef]
- 59. Mauro, R.; Branco, F. Two Vehicular Headways Time Dichotomic Models. Mod. Appl. Sci. 2012, 6, 1. [CrossRef]
- 60. Yin, S.; Li, Z.; Zhang, Y.; Yao, D.; Su, Y.; Li, L. Headway Distribution Modeling with Regard to Traffic Status. In Proceedings of the 2009 IEEE Intelligent Vehicles Symposium, Xi'an, China, 3–5 June 2009; IEEE: Piscataway, NJ, USA, 2009; pp. 1057–1062.
- 61. Buckley, D.J. A Semi-Poisson Model of Traffic Flow. *Transp. Sci.* **1968**, *2*, 107–133. [CrossRef]
- 62. Branston, D. Models of Single Lane Time Headway Distributions. Transp. Sci. 1976, 10, 125–148. [CrossRef]
- 63. Del Serrone, G.; Cantisani, G.; Peluso, P. Blending of Floating Car Data and Point-Based Sensor Data to Deduce Operating Speeds under Different Traffic Flow Conditions. *Eur. Transp. Trasp. Eur.* **2023**, *91*, 1–11. [CrossRef]
- 64. Osservatorio Del Traffico | Anas S.p.A. Available online: https://www.stradeanas.it/it/le-strade/osservatorio-del-traffico (accessed on 9 May 2023).
- Daniel, W.W. Kolmogorov-Smirnov One-Sample Test. In *Applied Nonparametric Statistics*, 2nd ed.; PWS-Kent Bost: Boston, MA, USA, 1990; pp. 319–330.
- 66. Johnson, R.A.; Wichern, D.W. Applied Multivariate Statistical Analysis; Prentice Hall: Hoboken, NJ, USA, 1996.
- 67. Rossi, R.; Gastaldi, M. An Empirical Analysis of Vehicle Time Headways on Rural Two-Lane Two-Way Roads. *Procedia Soc. Behav. Sci.* 2012, 54, 865–874. [CrossRef]
- 68. Lin, M.; Lucas, H.C.; Shmueli, G. Too Big to Fail: Large Samples and the p-Value Problem. *Inf. Syst. Res.* 2013, 24, 906–917. [CrossRef]
- Baird, G.L.; Harlow, L.L. Does One Size Fit All? A Case for Context-Driven Null Hypothesis Statistical Testing. J. Mod. Appl. Stat. Methods 2016, 15, 100–122. [CrossRef]
- Azevedo, H.V.F.; de Lima Junior, E.T. Statistical Inference Techniques Applied to Large Samples. In Proceedings of the XL Ibero-Latin-American Congress on Computational Methods in Engineering, Natal, Brazil, 11–14 November 2019.
- Rossi, R.; Gastaldi, M.; Pascucci, F.; Rossi, R.; Gastaldi, M.; Pascucci, F. Fitting Time Headway–Vehicle Speed Bivariate Distributions: Operational Procedure for Two-Way Two-Lane Roads. *Transp. Res. Rec.* 2015, 2490, 94–105. [CrossRef]
- 72. Martinelli, V.; Ventura, R.; Bonera, M.; Barabino, B.; Maternini, G. Effects of Urban Road Environment on Vehicular Speed: Evidence from Brescia (Italy). *Transp. Res. Procedia* **2022**, *60*, 592–599. [CrossRef]

- 73. Martinelli, V.; Ventura, R.; Bonera, M.; Barabino, B.; Maternini, G. Estimating Operating Speed for County Road Segments-Evidence from Italy. *Int. J. Transp. Sci. Technol.* **2023**, *12*, 560–577. [CrossRef]
- 74. Polus, A.; Mattar-Habib, C. New Consistency Model for Rural Highways and Its Relationship to Safety. *J. Transp. Eng.* **2004**, 130, 286–293. [CrossRef]
- 75. Camacho-Torregrosa, F.J.; Pérez-Zuriaga, A.M.; Campoy-Ungría, J.M.; García-García, A. New Geometric Design Consistency Model Based on Operating Speed Profiles for Road Safety Evaluation. *Accid. Anal. Prev.* **2013**, *61*, 33–42. [CrossRef] [PubMed]
- Morcillo, L.G.; Poyo, F.J.C.; Fernández, M.P.; de Oña, J. Measurement of Road Consistency on Two-Lane Rural Highways in Granada (Spain). *Procedia-Soc. Behav. Sci.* 2014, 162, 237–242. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.